Author's response to reviews

Title: Comprehensive assessment of phenolics and antioxidant potential of Rumex hastatus D. Don. roots

Authors:

Sumaira Sahreen (Sumairasahreen@gmail.com)
Muhammad R Khan (mrkhangau@yahoo.com)
Rahmat Khan (rahmatgul_81@yahoo.com)

Version: 4 Date: 5 June 2013

Author's response to reviews: see over
To
The Editor,
BMC Complementary and Alternative Medicine

Dear Sir,

Refer to “Comprehensive assessment of phenolics and antioxidant potential of *Rumex hastatus* D. Don. roots” according to your suggestions, I have revised the whole manuscript and added the required text. Hope that now the manuscript will be able for publication.

With regards

Dr. Rahmat Ali Khan
Assistant professor
Department of Biotechnology, UST Bannu
Email: rahmatgul_81@yahoo.com